\documentclass[10pt]{article} 
\usepackage{amssymb}
\usepackage{amsmath}
\newtheorem{example}{Example}
\newtheorem{lem}{Lemma}
\newtheorem{co}{Corollary}
\newtheorem{remark}{Remark}
\newtheorem{pr}{Proposition}
\newtheorem{no}{Notation}
\newtheorem{con}{Conjecture}
\newenvironment{proof}{\noindent {\em Proof:}}{\hspace*{1cm}
        \hspace*{\fill}$\rule{1.2ex}{1.4ex}$\medskip}
\newenvironment{re}{\begin{remark}\rm}{\end{remark}}
\newenvironment{ex}{\begin{example}\rm}{\end{example}}
\newenvironment{de}{\begin{definition}\rm}{\end{definition}}
\newcommand{\un}[1]{\underline{{\em #1}}}
\newcommand{\ebox}{\framebox[4mm]{\rule[0mm]{0mm}{2mm}}}
\newcommand{\xbox}{\framebox[4mm]{\rule[0mm]{0mm}{2mm}x}}


\begin{document}

Quiz, Math 110, Friday, March 19, 1999

\vskip.30 in
1.  Solve the following equation for $x$:
$
2 \ln{x} + \ln{16} = 5 \ln{2}
$\vskip.15 in
2.  Compute the derivative of the function:
$
f(x) = 2^{\ln{x}}.
$



\end{document}